**Author's response to reviews**

**Title:** Potential benefits of using a toolkit developed to aid in the adaptation of HTA reports: A case study considering positron emission tomography (PET) and Hodgkin's Disease

**Authors:**

- Sheila M Turner (s.turner@soton.ac.uk)
- Neil Adams (neil.adams@scpct.nhs.uk)
- Andrew Cook (andrewc@soton.ac.uk)
- Alison Price (alison@soton.ac.uk)
- Ruairidh Milne (rm2@soton.ac.uk)

**Version:** 2  **Date:** 28 October 2009

**Author's response to reviews:**

Page 14, transcription error E[22] should have been D[22]. Revision made.

Table 1, transcription errors in labelling of second row "pevious reports". Columns should have been labelled A - H. Revisions made in uploaded version.

Apologies for the mistakes. Thank you.